University of Massachusetts Amherst ScholarWorks@UMass Amherst

Masters Theses 1911 - February 2014

2012

A Theoretical Approach to Fault Analysis and Mitigation in Nanoscale Fabrics

Md Muwyid Uzzaman Khan University of Massachusetts Amherst

Follow this and additional works at: https://scholarworks.umass.edu/theses Part of the Other Electrical and Computer Engineering Commons, and the VLSI and Circuits, Embedded and Hardware Systems Commons

Khan, Md Muwyid Uzzaman, "A Theoretical Approach to Fault Analysis and Mitigation in Nanoscale Fabrics" (2012). Masters Theses 1911 - February 2014. 918.

Retrieved from https://scholarworks.umass.edu/theses/918

This thesis is brought to you for free and open access by ScholarWorks@UMass Amherst. It has been accepted for inclusion in Masters Theses 1911 -February 2014 by an authorized administrator of ScholarWorks@UMass Amherst. For more information, please contact scholarworks@library.umass.edu.

A THEORETICAL APPROACH TO FAULT ANALYSIS AND MITIGATION IN NANOSCALE FABRICS

A Thesis Presented

By

MD MUWYID UZZAMAN KHAN

Submitted to the Graduate School of the University of Massachusetts Amherst in partial fulfillment of the requirements for the degree of

MASTER OF SCIENCE IN ELECTRICAL AND COMPUTER ENGINEERING

SEPTEMBER 2012

Department of Electrical and Computer Engineering

© Copyright by Md Muwyid Uzzaman Khan 2012

All Rights Reserved

A THEORETICAL APPROACH TO FAULT ANALYSIS AND MITIGATION IN NANOSCALE FABRICS

A Thesis Presented

By

MD MUWYID UZZAMAN KHAN

Approved as to style and content by:

Csaba Andras Moritz, Chair

Israel Koren, Member

C. Mani Krishna, Member

C.V. Hollot, Department Head Department of Electrical and Computer Engineering

ACKNOWLEDGEMENTS

First and foremost, all praises and thanks are to Almighty Allah. I would also like to take this opportunity to thank all the people that helped me complete this thesis. I want to sincerely thank my advisor, Prof. Csaba Andras Moritz, for his generous advice, guidance and support. I would like to extend my gratitude to my committee members, Prof. Koren and Prof. Krishna for their time and effort in supporting my work. I would like to thank my fellow group members who provided immense intellectual as well as emotional support, especially Pritish Narayanan for his advice and guidance. Last but not the least, I would like to thank my parents, my sisters and the rest of my family and friends for their continuous support and motivation.

ABSTRACT

A THEORETICAL APPROACH TO FAULT ANALYSIS AND MITIGATION IN NANOSCALE FABRICS

September 2012

MD MUWYID UZZAMAN KHAN B.Sc., ISLAMIC UNIVERSITY OF TECHNOLOGY M.S.E.C.E., UNIVERSITY OF MASSACHUSETTS AMHERST

Directed by: Professor Csaba Andras Moritz

High defect rates are associated with novel nanodevice-based systems owing to unconventional and self-assembly based manufacturing processes. Furthermore, in emerging nanosystems, fault mechanisms and distributions may be very different from CMOS due to unique physical layer aspects, and emerging circuit and logic styles. Thus, theoretical fault models for nanosystems are necessary to extract detailed characteristics of fault generation and propagation. Using the intuition garnered from the theoretical analysis, modular and structural redundancy schemes can be specifically tailored to the intricacies of the fabric in order to achieve higher reliability of output signals.

In this thesis, we develop a detailed analytical fault model for the Nanoscale Application Specific Integrated Circuits (NASIC) fabric that can determine probabilities of output faults taking into account the defect scenarios, the logic and circuit style of the fabric as well as structural redundancy schemes that may be incorporated in the circuits. Evaluation of fault rates using the analytical model for single NASIC tiles show an inequality of the probability of output faulty '1's and '0's. To mitigate the effects of the unequal fault rates, biased voting schemes are introduced and are shown to achieve up to 27% improvement in the reliability of output signals compared to conventional majority voting schemes.

NASIC circuits have to be cascaded in order to build larger systems. Furthermore, modular redundancy alone will be insufficient to tolerate high defect rates since multiple input modules may be faulty. Hence incorporation of structural redundancy is crucial. Thus in this thesis, we study the propagation of faults through a cascade of NASIC circuits employing the conventional structural redundancy scheme which is referred to here as the Regular Structural Redundancy. In our analysis we find that although circuits with Regular Structural Redundancy achieve greater signal reliability compared to nonredundant circuits, the signal reliability rapidly drops along the cascade due to an escalation of faulty '0's. This effect is attributed to the poor tolerance of input faulty '0's exhibited by circuits with the Regular Structural Redundancy. Having identified this, we design a new scheme called the *Staggered Structural Redundancy* prioritizing the tolerance of input faulty '0's. A cascade of circuits employing the Staggered Structural *Redundancy* is shown to maintain signal reliability greater than 0.98 for over 100 levels of cascade at 5% defect rate whereas the signal reliability for a cascade of circuits with the Regular Structural Redundancy dropped to 0.5 after 7 levels of cascade.

TABLE OF CONTENTS

Page

	-
ACKNOWLEDGEMENTS	iv
ABSTRACT	v
LIST OF TABLES	ix
LIST OF FIGURES	X
1. INTRODUCTION	1
2. RELATED BACKGROUND	4
2.1 Overview of NASICs	4
2.2 Fault Tolerance in Nanofabrics	5
2.3 Structural Redundancy in NASICs	5 6
2.5 Structural Redundancy in 1945105	0
3 ANALVTICAL FAILT MODEL FOR NASICS	8
3.1 Introduction	0 و
3.1 Introduction	0
3.2 Defect Wodel	0
3.3 Assumptions	9
	9
3.5 Thes without Structural Redundancy	11
3.5.1 Occurrence of Faulty 1 ⁷	11
3.5.2 Occurrence of Faulty '0'	15
3.6 Tiles with Regular Structural Redundancy	16
3.6.1 Occurrence of Faulty '1'	17
3.6.2 Occurrence of Faulty '0'	18
3.7 Summary	19
4. BIASED VOTING FOR IMPROVED SIGNAL RELIABILITY	20
4.1 Introduction	20
4.2 Fault Probability Ratio	21
4.3 Biased Voting in General Case	24
4.4 Biased Voting in NASICs	30
4.5 Voter Reliability	33
4.6 Summary	36
5. STRUCTURAL REDUNDANCY IN CASCADED NASICS	37
5.1 Introduction	37
5.2 Regular Structural Redundancy	38
5.2.1 Fault Propagation along cascade	38
5.2.2 Resilience to Input Faults	44
5.3 Staggered Structural Redundancy	45
5.3.1 Introduction	
5.3.2 Criteria and Notation	
5.3.3 Simulation Framework	48
5.3.4 Resilience to Input Faults	+0 48
5 3 5 Fault Pronagation along caseade	7 0 51
5.4 Fyemple: Rinple Cerry Adder	51 51
5.5 Example: Cascada of 2 bit Comparators	
5.6 Songitivity to Stude Off Defeats	30 50
	30

5.7 Summary	
6. CONCLUSION	61
BIBLIOGRAPHY	

LIST OF TABLES

Table	Page
Table 1. Notations	10
Table 2. Notations	25
Table 3: Matrix notation of structural redundancy schemes	47
Table 4: Cascade level after which Signal Reliability is 0.5	54

LIST OF FIGURES

Figure Pa	ıge
Figure 1. A full adder implemented in NASICs	. 5
Figure 2. NASIC fault tolerance schemes: (a) NASIC tile with no redundancy incorporated. (b) 2-way redundant implementation of the NASIC tile in (a)	.7
Figure 3. An n-input NASIC tile built with 2 NAND stages	11
Figure 4. Fault Probability Ratio as a function of number of minterms for n=8 at defect rates of 5%, 10% and 15%	23
Figure 5. Fault Probability Ratio as a function of defect rate for a range of fan-in (n) when $m=2^{n-1}$	23
Figure 6. Voting configurations: (a) TMR majority voting, (b) $V_0^{1/3}$ biased voting	24
Figure 7. Signal reliability as a function of Fault Probability Ratio (FPR) for (a) $P0/1 = 0.05$, $P(V^{C}=0) = 0.5$ and (b) $P0/1 = 0.1$, $P(V^{C}=0) = 0.5$ (c) $P0/1 = 0.1$, $P(V^{C}=0) = 0.75$ and (d) $P0/1 = 0.1$, $P(V^{C}=0) = 0.25$	29
Figure 8. Signal Reliability comparison for biased and majority voting schemes at defect rates of: (a) 5%, (b) 10%, and (c) 15%	32
Figure 9. $V_0^{2/4}$ biased voter in NASICs	33
Figure 10. Signal Reliability comparison for biased and majority voting schemes after considering voter defects at defect rates of: (a) 5%, (b) 10%, and (c) 15%	35
Figure 11: Regular Structural Redundancy schemes for a NASIC tile implementing a 2- input XOR logic function: (a) NASIC tile with no structural redundancy incorporated (b) NASIC tile with 2-way RSR	38
Figure 12: Cascade structure used to analyze propagation of faults	39
Figure 13: Probability of output faulty '0' and faulty '1' and Signal Reliability of cascaded tiles with n=4 and m=8 at 10% stuck-on probability for (a) tiles with no structural redundancy (b) with 1w-RSR and (c) with 2w-RSR	40
Figure 14: 3w-RSR tile demonstrating diminished resilience to input faulty '0'	42
Figure 15: Resilience to input faulty '0' by non-redundant tile, 2w-RSR and 3w-RSR for n=4, m=8. (a) 4-input tile with 10% defect rate and with faulty input '0's. (b) Probability of output faults as a function of probability of input faulty '0's	44
Figure 16. (3,2) staggered structural redundancy of 2-input XOR gate in NASICs	46

Figure 17: Resilience to input faulty '0' by 3w-RSR and (3,2)-SSR for n=4, m=8. (a) 4-input XOR tile with 10% defect rate and with faulty input '0's. (b)	
Probability of output faults for the two schemes (c) Signal Reliability for the two schemes	49
Figure 18:(a)Probability of output faults and (b) Signal Reliability along cascade of tiles	
with n=4 and m=8 at 10% detect rate	52
Figure 19: Signal Reliability along cascade of 2w-RSR, 3w-RSR and (3,2)-SSR for tiles with n=4, m=8 and at (a) 5% defect rate and (b) 15% defect rate	53
 Figure 20: (a) 16-bit Ripple carry adder with ideal inputs and faulty carry signal. (b) Signal Reliability of carry signal at each cascade level for 3w-RSR and (3,2)- SSR at 10% defect rate (b) Signal Reliability of the (3,2)-SSR at 5%, 10%, 15% and 20% defect rates 	55
Figure 21: (a) Cascade of 2-bit comparators. (b) Signal Reliability along cascade of 3w- RSR and (3,2)-SSR implementations at 5%, (b) 10% and (c) 15% defect rate	57
Figure 22: Signal Reliability along cascade with 10% total defect rate and PS-OFF stuck- off rate for (a) 3w-RSR and (b) (3,2)-SSR	60

CHAPTER 1 INTRODUCTION

A variety of nano-materials and nano-devices including semiconductor nanowires [1],[2], carbon nanotubes [3], quantum cellular automata (QCA) [4], graphene-based devices [5], spin-wave fabrics [6] and molecular devices [7] have been proposed as alternatives to conventional CMOS. However, self-assembly and unconventional manufacturing approaches for scalable assembly of nanostructures may imply orders of magnitude higher defect rates compared to CMOS [8]. Furthermore, nanoscale computational fabrics may have very different fault models due to novel circuit and logic styles and different defect scenarios. It is thus necessary to first study the fault characteristics in nanoscale fabrics using detailed theoretical models. The intuition garnered from this analysis will facilitate the design of fault-tolerance schemes that are better tailored to these unique fault characteristics thus achieving improved yield.

In this thesis, we derive detailed analytical expressions for the probability of '0' to '1' faults (referred to as faulty '1's) and '1' to '0' faults (faulty '0's) at the output of circuits implemented in the Nanoscale Application Specific ICs (NASICs) fabric [9] [10]. The expressions are in terms of the probability of defects, input faults and the specific logic and circuit styles of the fabric. The detailed analytical fault models are also capable of accounting for structural redundancies that may be implemented within the circuits. Results obtained from analyzing single NASIC tiles using the model indicate that the probabilities of the two types of faults are unequal. Since one fault is more likely to occur, biased voting schemes instead of conventional majority voting schemes provide higher signal reliabilities by offering greater protection against the most likely faults. Our theoretical analysis shows that the biased voting schemes provide up to 27% improvement in signal reliability compared to the conventional majority voting schemes.

To build larger systems NASIC circuits will have to be cascaded. Hence, evaluation of the fault characteristics and improvement of the signal reliability through cascaded NASIC tiles is necessary. Furthermore, modular redundancy alone will not be sufficient to tolerate the high defect rates characteristic of nanoscale fabrics since multiple input modules may be faulty. Thus, structural redundancy needs to be incorporated within individual circuits. Thus in this thesis, we study the propagation of faulty '1's and faulty '0's through cascades of structurally redundant NASIC circuits and their effect on the overall Signal Reliability. In our analysis, we demonstrate that the Signal Reliability through cascades of circuits employing the conventional structural redundancy scheme (referred to as Regular Structural Redundancy (RSR)) rapidly drops along the cascade due to an escalation of the probability of faulty '0's. Having identified that Regular Structural Redundancy schemes are intolerant of input faulty '0's, we propose and introduce a novel structural redundancy scheme called the Staggered Structural Redundancy (SSR) that offers improved resilience against input faulty '0's. The novel Staggered Structural Redundancy schemes are shown to have 56% improved tolerance of input faulty '0's compared to the Regular Structural Redundancy. Our results show that although the signal reliability through cascades of RSR circuits drops to 0.5 after 7 levels of cascades at 10% defect rate, the signal reliability through cascades of SSR circuits remain above 0.98 for over 100 levels of cascade due to its adequate capability of tolerating both input faulty '1's and '0's.

The rest of the thesis is organized as follow. Chapter 2 discusses related prior work in the fault tolerance in nanofabrics and provides overview of NASICs and its structural redundancy schemes. Chapter 3 provides the derivation of the detailed analytical fault models for the NASIC fabric, both for circuits with and without structural redundancy. Chapter 4 utilizes the analytical models to demonstrate the benefits of biased voting schemes in improving signal reliability. Chapter 5 provides a study of the fault propagation through cascades of circuits with conventional structural redundancy and introduces a novel structural redundancy which is shown to achieve improved signal reliability. Finally, chapter 6 summarizes this thesis.

CHAPTER 2 RELATED BACKGROUND

2.1 Overview of NASICs

Nanoscale Application Specific Integrated Circuits (NASICs) [11]-[16] is a computational fabric based on a 2-dimensional grid of semiconductor nanowires with external dynamic control for data streaming and cascading.

Cross-nanowire transistors (xnwFETs) are formed at selected cross-points to implement the logic function. Microwires are used to provide VDD, GND and control signals for data streaming. In NASICs, the 2-stage dynamic NAND-NAND logic style is one of the logic families used [14]. The output signals from the first stage NAND gates become the input signals for the nanowire transistors in the second stage NAND gate. Dynamic circuits and pipelining in NASICs obviate the need for explicit latching and improves the density. Furthermore, the NAND-NAND logic style requires only n-type xnwFETs leading to a simplified manufacturing pathway.

The elemental unit in NASICs is called a *Tile*. It is a single NASIC circuit implementing a 2-stage logic function. Figure 1 shows a single NASIC tile (consisting of 2 dynamic NAND stages) implementing a 1-bit full adder. In Figure 1, hpre and heva are the precharge and evaluate control transistors that enable dynamic circuit evaluation. Many such tiles can be cascaded together to build a large-scale system such as a processor [17] or an image processing architecture [18].



Figure 1. A full adder implemented in NASICs

In NASICs, high fan-in circuits are possible since delay scales linearly with respect to fan-in as opposed to conventional CMOS where the trend is typically quadratic [19]. This is due to the unique dynamic control schemes used, where successive cascaded stages are evaluated using different control signals. The series stack resistance of a given stage is overcome during and after the pre-charge of the previous stage. This implies that during the evaluation of the current stage, only the linear impact of capacitance affects the performance with increasing fan-in. This behavior has been verified through detailed simulations of device behavior and circuit characteristics. Additional details can be found in [19].

2.2 Fault Tolerance in Nanofabrics

Self-assembly and unconventional manufacturing approaches for scalable assembly of nanostructures may imply orders of magnitude higher defect rates compared to CMOS [8]. Therefore, built-in fault tolerance techniques need to be incorporated in nanoscale systems at various levels to achieve high-yield systems. Prior work on fault tolerance of hard defects includes techniques such as reconfiguration and built-in redundancy. Techniques for reconfigurable fabrics include mapping of logic functions onto defective circuits or reconfiguring around defective blocks [20] -[23] and built-in self test techniques for testing and diagnosis [24][25]. In hardware redundancy techniques for non-reconfigurable crossbar architectures, a lot of the previous work concentrated on the efficient mapping of logic functions onto defective crossbars [26]-[28]. Both these approaches lead to technical challenges such as the need for special reconfigurable devices or the complex interfacing between micro and nano circuits to extract defect maps.

Modular redundancy techniques have been widely researched in the past few decades and include Triple modular redundancy (TMR) [29]-[32] and N-tuple modular redundancy [33]. Designs of new voter circuits [34] [35] and even NMR systems without a centralized voter [36] have been proposed. A more fine-grained built-in fault tolerance technique is the structural redundancy [37]. As the focus shifts to nanoscale devices, hardware redundancy techniques still hold promise [38]-[40].

2.3 Structural Redundancy in NASICs

In NASICs, possible defects include broken nanowires, stuck-on and stuck-off type defects. Reliable manufacturing of nanowires up to a few microns in length has been demonstrated in [2], so the frequency of broken nanowires is expected to be very low. Stuck-on transistors are the most prevalent in the proposed manufacturing pathway [41] due to the ion implantation and metallization processes involved.



Figure 2. NASIC fault tolerance schemes: (a) NASIC tile with no redundancy incorporated. (b) 2-way redundant implementation of the NASIC tile in (a).

Structural redundancy is employed in NASICs by creating redundant copies of nanowires. Redundant signals are created and logically merged in the logic planes with regular signals. Every xnwFET has a redundant copy so that even if one of the copies is stuck-on, then the redundant copy or copies will be able to switch off, thus ensuring proper logic implementation.

Figure 2 (a) shows a single non-redundant NASIC tile and Figure 2 (b) shows its 2-way redundant implementation. In the 2-way redundant tile, every nanowire is duplicated, each containing twice as many xnwFETs. The redundant signals are merged within the logic plane itself. In Figure 2 (a) and (b), $\overline{a_0}$ and a'_0 is the complement and the redundant signal of a_0 , respectively.

CHAPTER 3

ANALYTICAL FAULT MODEL FOR NASICs

3.1 Introduction

Analytical fault models for NASICs are necessary to extract detailed characteristics of error generation and propagation. This will facilitate the exploration of improved fault tolerance schemes that are specifically tailored to the intricacies of the fabric. In this chapter, we first derive an analytical fault model for a NASIC tile without any structural redundancy and then extend it to include tiles with Regular Structural Redundancy. The analytical model is capable of calculating output fault rates and signal reliabilities in the presence of stuck-on defects and input faults.

3.2 Defect Model

Defects in the NASICs fabric depend on the manufacturing pathway used. One possible manufacturing pathway has been described in [41]. Reliable manufacturing of nanowires up to a few microns in length has been demonstrated in [2], so the frequency of broken nanowires is assumed to be negligible. Stuck-on transistors are the most prevalent in this pathway due to the ion implantation and metallization processes involved. Thus in the derivation of the model only stuck-on type defects are considered. The probability of a transistor being stuck-on, denoted by P_{S-ON} , thus represents the defect rate of the NASIC fabric in this chapter. These defects are considered to occur independently of each other, since they are caused by local effects (e.g. lateral diffusion after ion implant).

A defect rate of up to 15% is considered because, according to our initial work [13], at defect rates higher than this, any density advantage over projected CMOS would likely be eliminated in the context of microprocessor designs. It must be noted that this is a device-level defect rate and is 10 orders of magnitude higher than in scaled CMOS. For instance, CMOS defect rates are 0.4 defects/cm² [42] whereas 1%-15% defect rate in NASICs translates to billions of defects/cm².

3.3 Assumptions

In the derivation of the model, stuck-on defects in the NASIC tile are assumed to occur independently since they are caused by local effects such as lateral diffusion after ion implantation. The following assumptions are also made for tractability of the model:

- Logic functions on the NASIC tiles are assumed to be implemented in the canonical sum of products form without minimization.
- All the inputs of a tile are assumed to have the same probability of being faulty and they are assumed to occur independently.
- Occurrences of faults on horizontal nanowires are assumed to be independent of one another.

3.4 Notations

Figure 3 shows two NAND stages in a single n-input NASIC tile. Here, input signals are denoted by *i*. M_0 to M_{m-1} are the minterms generated by the first NAND stage and $T2_0$ to $T2_{m-1}$ denote the transistors in the 2nd stage NAND gate.

The rest of the notations that have been used in the derivation of the model are given in Table 1.

Table 1. Notations

Symbol	Description
n	Number of inputs for the logic function implemented
m	Number of minterms generated by the first NAND stage
S ^C	Correct output of logic function
S ^A	Actual fault-prone output from a defective NASIC tile
M _i ^C	Correct ith minterm expected from a defect-free circuit, $0 \le i \le m-1$
Mi ^A	Actual fault-prone ith minterm in a defective circuit
T1 ^C (T1 ^A)	Correct (actual) state of Transistor in the first NAND stage
$T2_i^C (T2_i^A)$	Correct (actual) state of Transistor gated by minterm Mi in the second stage NAND gate
$P_{0/1}^{out} =$ $Prob\{S^{A}=0 S^{C}=1\}$	Probability of a faulty '0' at the output of a defective tile
$P_{1/0}^{out} =$ $Prob\{S^{A}=1 S^{C}=0\}$	Probability of a faulty '1' at the output of a defective tile
$P_{0/1}^{min} =$ $Prob\{M_i^{A}=0 M_i^{C}=1\}$	Probability of the minterm, Mi, being a faulty '0'
$P_{1/0}^{min} =$ Prob{ $M_i^A = 1 M_i^C = 0$ }	Probability of the minterm, Mi, being a faulty '1'
P _{0/1}	Probability that an input is faulty '0'

$P_{1/0}^{inp}$	Probability that an input is faulty '1'
	$Prob(T1_i^A = ON T1_i^C = OFF)$; Probability that
$P_{ON/OFF}^{T1}$	transistor in first NAND stage, 11, is switched-on due to fault whereas ideally it should have been switched-
	off
	$Prob(T2_i^A = ON T2_i^C = OFF)$; Probability that
DT2	transistor in second NAND stage, T2, is switched-on
^I ON/OFF	switched-off
P _{S-ON}	Probability that a transistor is Stuck-ON



Figure 3. An n-input NASIC tile built with 2 NAND stages.

3.5 Tiles without Structural Redundancy

3.5.1 Occurrence of Faulty '1'

In this section we derive the equation for the probability of occurrence of a faulty '1' (0-to-1 fault) at the output of a structurally non-redundant NASIC tile, given the probability of stuck-on defects, stuck-off defects and input faults.

In a 2-stage dynamic NAND-NAND logic implementation, a '0' is produced at the output if all of the input signals (minterms) to the second stage NAND gate are '1's. Thus, in a NASIC tile, all of the xnwFETs (transistors) in the second stage dynamic NAND gate must be correctly switched on to allow the output to evaluate to '0'. Hence, for a faulty '1' to be produced at the output, any one of the transistors has to be switchedoff to prevent evaluation to '0'. This will happen if any of the transistors in the 2nd stage NAND gate is incorrectly switched-off. Mathematically,

$$P_{1/0}^{out} = Prob\{S^{A} = 1 \mid S^{C} = 0\} = 1 - Prob\{S^{A} = 0 \mid S^{C} = 0\}$$
(3.1)

$$P_{1/0}^{out} = 1 - Prob\{\bigcap_{i=0}^{m-1} (T2_i^A = ON | T2_i^C = ON)\}$$
(3.2)

A transistor, $T2_i$ will be correctly switched-on in the event that the transistor is functional and the minterm gating it carries a correct '0', or in the event that the transistor is simply stuck-on. Thus,

$$P_{ON/ON}^{T2} = Prob\{T2_i^A = ON \mid T2_i^C = ON\}$$

$$P_{ON/ON}^{T2} = Prob\{T2_i = stuck - on \cup (T2_i = not stuck on \cap M_i^A = 1|M_i^C = 1)\}$$
(3.3)

$$P_{ON/ON}^{T2} = P_{S-ON} + (1 - P_{S-ON})(1 - P_{0/1}^{min})$$
(3.4)

Substituting Equation 3.4 into Equation 3.2 and simplifying, we get an expression for the probability of occurrence of a faulty '1' at the output of a NASIC tile,

$$P_{1/0}^{out} = 1 - (1 - P_{0/1}^{min} + P_{S-ON} \times P_{0/1}^{min})^m$$
(3.5)

The above probability of a faulty '1' has been expressed in terms of the probability of a transistor being stuck-on, P_{S-ON} , the number of minterm signals, m, and the probability of a minterm producing an incorrect '0', $P_{0/1}^{min}$.

In this model, the probability of a minterm being a faulty '0', $P_{0/1}^{min}$, is entirely dependent on the transistors of the first stage NAND gates being defective. The transistors in the first NAND stage are gated by tile inputs. For an n-input NASIC tile, there are 2ⁿ possible input combinations. Since each of the horizontal nanowires in the first stage is a dynamic NAND gate, only for one particular input pattern should the output of the horizontal nanowire be '0'. The output should be a '1' for all of the rest of the input patterns. For each of these input patterns that should produce an output '1', one or more transistors will be correctly switched off to keep the output of the minterm at '1'. Thus, for a minterm to be a faulty '0', one or more transistors need to be incorrectly switched-on (depending on the input pattern), and the rest of the transistors need to remain correctly switched-on. For instance, in a 2 input logic function implemented in the NASIC fabric, at any one of the first stage NAND gates, there are 3 input patterns $(2^2 - 1 = 3)$ that should produce an output of '1' at that NAND gate. 1 out of those 3 input patterns will require both transistors in the gate to be incorrectly switched-on to produce a faulty '0'. 2 of the input patterns will require 1 transistor to be incorrectly

switched-on and the other transistor to be remain correctly switched-on. Assuming all input patterns are equally probable, $P_{0/1}^{min}$ for this specific case of 2-input logic function would be,

$$P_{0/1}^{min} = \frac{1}{3} \left(P_{ON/OFF}^{T1}^{2} + 2P_{ON/OFF}^{T1} P_{ON/ON}^{T1} \right)$$
(3.6)

Extending this for an n-input NASIC tile and using the binomial theorem for simplification, the probability of minterm faulty '0' in an n-input tile is,

$$P_{0/1}^{min} = \frac{1}{2^n - 1} \left[\left(P_{ON/OFF}^{T1} + P_{ON/ON}^{T1} \right)^n - P_{ON/ON}^{T1} \right]$$
(3.7)

The above probability is written in terms of the probability of transistors in the first NAND stage being correctly and incorrectly switched-on. A transistor in the first stage may be incorrectly switched-on if it is either stuck-on or if the tile input gating the transistor is carrying a faulty '1'. Thus, the probability of T1 being incorrectly switched-on can be written as,

$$P_{ON/OFF}^{T1} = P_{S-ON} + (1 - P_{S-ON})(P_{1/0}^{inp})$$
(3.8)

Similarly, the probability of a transistor in the 1st NAND stage being correctly switched-on is the probability of the event that the transistor is stuck-on or the transistor is functional with the input gating the transistor carrying a correct logic '1'. Thus,

$$P_{ON/ON}^{T1} = P_{S-ON} + (1 - P_{S-ON})(1 - P_{0/1}^{inp})$$
(3.9)

Finally, by substituting Equation 3.8 and 3.9 into Equation 3.7 and subsequently substituting Equation 3.7 into Equation 3.5, the expression for the probability of an

output faulty '1' is expressed in terms of the probability of stuck-on defects, the probability of input faults, the fan-in and the number of minterms of the specific NASIC tile.

3.5.2 Occurrence of Faulty '0'

In this section we derive the equation for the probability of occurrence of a faulty '0' (1-to-0 fault) at the output of a NASIC tile. A faulty '0' is said to occur when the inputs to a tile are such that in an ideal case, the output of the tile should be a '1' but due to defects within the tile and input faults from the preceding tile, the output is a '0'.

When an input pattern that should produce a '1' at the output of a defect-free circuit arrives at such a 2-stage NAND-NAND circuit, all of the minterms (inputs to the second NAND stage) carry logic '1' except for one minterm, Mx, that carries a logic '0'. Thus, only one transistor, say Tx, at the second stage NAND gate is correctly switched-off in a defect-free circuit to keep the output at logic '1'. Hence, for a faulty '0' to occur in a defective circuit, transistor Tx should be incorrectly switched-on while the other transistors remain switched-on, enabling evaluation to faulty '0'.

$$P_{0/1}^{out} = Prob(S^A = 0|S^C = 1)$$
(3.10)

$$P_{0/1}^{out} = Prob\{(\bigcap_{i=0; i \neq x}^{m-1} T2_i^A = ON | T2_i^C = ON) \cap (T2_x^A = ON | T2_x^C = OFF)\}$$
(3.11)

$$P_{0/1}^{out} = P_{ON/ON}^{T2} \times P_{ON/OFF}^{T2}$$
(3.12)

Transistor $T2_x$ will incorrectly switch-on in the event that the transistor is functional but the minterm gating the transistor is incorrectly '1' or in the event that the transistor is stuck-on.

$$P_{ON/OFF}^{T2} = P_{S-ON} + (1 - P_{S-ON})(P_{1/0}^{min})$$
(3.13)

The minterm will be a faulty '1' if at least one of the 1st stage transistors is incorrectly switched-off. Thus,

$$P_{1/0}^{min} = 1 - (P_{ON/ON}^{T1})^n \tag{3.14}$$

The above probability of minterm faulty '0' can be calculated by using Equation 3.9. By substituting Equation 3.4 and 3.13 into Equation 3.12, the probability of occurrence of a faulty '0' can be written as,

$$P_{0/1}^{out} = (P_{S-ON} + P_{1/0}^{min} - P_{S-ON} \times P_{1/0}^{min})$$

$$\times (1 - P_{0/1}^{min} + P_{S-ON} \times P_{0/1}^{min})^{m-1}$$
(3.15)

By substituting Equation 3.7 and 3.14 into Equation 3.15, probability of faulty '0' at the output of a NASIC tile with no structural redundancy can be obtained

3.6 Tiles with Regular Structural Redundancy

In this section we will extend the analytical model to incorporate α -way Regular Structural Redundancy. An overview of structural redundancy has been provided in Chapter 1. For tractability of the model, we assume that horizontal nanowires have no logical dependencies between them. The notation used for the probability of faulty output and minterm for a tile with α -way Regular Structural Redundancy is $P_{i/j}^{out-\alpha w}$ and $P_{i/j}^{min-\alpha w}$.

3.6.1 Occurrence of Faulty '1'

In tiles with structural redundancy, copies of are made of transistors on each nanowire and of the nanowires themselves. In a 2-way redundant tile for instance, every transistor will have a redundant copy in the same nanowire and every nanowire will be duplicated and logically merged with the next stage. Thus, in a 2-way redundant tile, twice as many transistors have to be correctly switched-on in order to produce a correct '0' at the output. Hence Equation 3.5 can be extended for the case of 2-way and α -way Regular Structural Redundancy as follows,

$$P_{1/0}^{out-2w} = 1 - (1 - P_{0/1}^{min-2w} + P_{S-ON} \times P_{0/1}^{min-2w})^{2m}$$
(3.16)

$$P_{1/0}^{out-\alpha w} = 1 - (1 - P_{0/1}^{min-\alpha w} + P_{S-ON} \times P_{0/1}^{min-\alpha w})^{\alpha m}$$
(3.17)

Note that the probability of a minterm being faulty '0' in a structurally redundant tile, $P_{0/1}^{\min-\alpha w}$, is not the same as the probability of a faulty '0' minterm in a non-redundant tile. This is because redundant copies of transistors are present on each horizontal nanowire gated by redundant copies of inputs. Thus, In order for a minterm faulty '0' to occur in 2-way structurally redundant tile for a specific input pattern, twice as many 1st stage transistors have to be incorrectly switched-on and twice as many transistors have to remain correctly switched-on compared to a non-redundant tile. Thus Equation 3.7 can be extended for a structurally redundant tile as follows,

$$P_{0/1}^{min-\alpha w} = \frac{1}{2^n - 1} \left[\left(P_{ON/OFF}^{T1}{}^{\alpha} + P_{ON/ON}^{T1}{}^{\alpha} \right)^n - P_{ON/ON}^{T1}{}^{\alpha n} \right]$$
(3.18)

Equations 3.8 and 3.9 remain valid for tiles with structural redundancy.

3.6.2 Occurrence of Faulty '0'

The probability of a faulty '0' for a structurally redundant tile can be extended from the non-redundant case in a similar manner as faulty '1'. Since there are α times the number of transistors in a α -way FSR compared to a non-redundant tile, the probability of a faulty '0' is,

$$P_{0/1}^{out-\alpha w} = P_{ON/ON}^{T2} \stackrel{\alpha(m-1)}{\times} P_{ON/OFF}^{T2} \stackrel{\alpha}{\longrightarrow} (3.19)$$

The expression for $P_{ON/OFF}^{T2}$ is the same as in Equation 3.13 except that the probability of a minterm being faulty '1' is different in the case of a structurally redundant tile since there are redundant transistors present in the horizontal nanowires.

$$P_{ON/OFF}^{T2} = P_{S-ON} + (1 - P_{S-ON})(P_{1/0}^{min-\alpha w})$$
(3.20)

Since there are α times the number of transistors in the first stage horizontal nanowire of a α -way FSR compared to a non-redundant tile, the probability of minterm faulty '1' can be expressed as,

$$P_{1/0}^{min-\alpha w} = 1 - (P_{ON/ON}^{T1})^{\alpha n}$$
(3.21)

Similarly $P_{ON/ON}^{T2}$ for α -way FSR is similar to Equation 3.4 with the probability of faulty minterm being different,

$$P_{ON/ON}^{T2} = P_{S-ON} + (1 - P_{S-ON})(1 - P_{0/1}^{min-\alpha w})$$
(3.22)

By substituting Equation 3.20 and 3.22 into Equation 3.19, the final expression for the probability of an output faulty '0' can be written as,

$$P_{0/1}^{out-\alpha w} = (P_{S-ON} + P_{1/0}^{min-\alpha w} - P_{S-ON} \times P_{1/0}^{min-\alpha w})^{\alpha}$$

$$\times (1 - P_{0/1}^{min-\alpha w} + P_{S-ON} \times P_{0/1}^{min-\alpha w})^{\alpha(m-1)}$$
(3.23)

3.7 Summary

In this chapter, we have derived detailed analytical fault models for NASIC tiles with and without Regular Structural Redundancy, taking into consideration the defect scenarios, input faults, circuit and logic style of the fabric. These models will be used in the next Chapters to analyze the relative proportions of the probability of faulty '1's and '0's and discuss and introduce both modular redundancy and structural redundancy techniques for their mitigation.

CHAPTER 4

BIASED VOTING FOR IMPROVED SIGNAL RELIABILITY

4.1 Introduction

Modular redundancy techniques have been widely researched in the past few decades and include Triple modular redundancy (TMR) [29]-[32] and N-tuple modular redundancy [33]. Designs of new voter circuits [34][35] and even NMR systems without a centralized voter [36] have been proposed.

In modular redundancy techniques, identical replicas of modules are created whose outputs are then voted upon. For instance, in TMR, the outputs of three identical copies of a (non-redundant) module are fed into a majority voter. If, for example, at least two of the three modules produce logic '0's, then the voter outputs a '0'. Otherwise, the voter outputs logic '1'. The inherent assumption that is made, when majority voters are employed, is that a '0' to '1' fault (faulty '1') is as likely as a '1' to '0' fault (faulty '0'). In other words, it is assumed that defects within the computational fabric may cause faulty '0's and faulty '1's with equal probability. However, with nanoscale computation fabrics based on novel circuit and logic styles, and different defect scenarios due to unconventional manufacturing, this assumption may no longer be valid. One type of fault may be more frequent than another. This provides an opportunity to potentially achieve higher yields by using biased voters, i.e., by offering greater protection against the most likely faults and less against the others.

In this chapter, we use the analytical models derived in the previous chapter to analyze the proportions of the probability of output faulty '1' and output faulty '0' for a structurally non-redundant NASIC tile with non-faulty inputs. We show that for such a tile the probability of output faulty '1's is greater than the probability of output faulty '0's. In other words, faulty '1's are more likely to occur at the output of such a tile compared to faulty '0's. Thus, we compare biased voters to traditional majority voting in such cases where the probabilities of faulty '0's and faulty '1's are unequal. We show that as the imbalance between the two fault probabilities is increased, the biased voting schemes become more effective. In section 4.3, we compare biased voting schemes to conventional majority voting schemes for general case without regard to any specific fabric. In section 4.4, we apply biased voting schemes to NASICs and analyze improvements in the reliability of outputs. Section 4.5 studies the effects on the results when the voters themselves are allowed to be defective. Section 4.6 summarizes the chapter.

4.2 Fault Probability Ratio

In order to study the relative proportions of the two types of faults for which probabilities were derived in the previous sections, we define a parameter called the Fault Probability Ratio (FPR). The Fault Probability Ratio is the ratio of the probability of a faulty '1' to the probability of a faulty '0'. Thus,

$$FPR = P_{1/0}^{out} / P_{0/1}^{out}$$
(4.1)

By substituting Equation 3.5 and Equation 3.15 into the above equation, the FPR can be calculated. From the equations derived in the previous sections it is evident that the FPR is a function of the defect probability, P_{S-ON} , the probability of input fault, $P_{i/i}^{inp}$,

the number of inputs of the logic function, n and the number of minterms, m. However, in this chapter the inputs to the tile are assumed to be non-faulty. Thus, $P_{0/1}^{inp} = P_{1/0}^{inp} = 0$.

Figure 4 shows the FPR of a structurally non-redundant NASIC tile with a fan-in of n = 8 for an increasing number of minterms (m) at three different defect rates. It is observed that the FPR rises with increasing number of minterms. This is expected since an increase in the number of transistors in the 2nd stage NAND gate makes faulty '0's less likely and faulty '1's more likely. A qualitative explanation for this is that a single transistor being incorrectly switched-off is sufficient to prevent the output node from discharging to '0'.

Furthermore, for a certain NASIC tile, the FPR is greater for a higher value of the defect rate, P_{S-ON} . For instance, the FPR is 8.1 for a tile with n = 8 and m = 160 at a defect rate of 5% and 18.22 for a defect rate of 15%. This can be better observed from Figure 5 that shows the FPR for a range of defect rates up to 15%. This trend occurs because as more transistors in the first NAND stage are stuck-on, the probability of the minterms carrying faulty '0's rises and thus there is an increased chance of a transistor in the second NAND stage gate being incorrectly switched-off and causing a faulty '1' at the output of the tile.

Defect rates of up to 15% are considered because any density advantage over CMOS is lost beyond this rate. In the plots of Figure 5, the average case (i.e., a symmetric function) was considered where the logic function implemented should produce (in the defect-free case) a '1' output for half of the 2^n input patterns and '0' for the remaining half. Hence, the number of minterms for such a logic function is, $m = 2^n - 1$.



Figure 4. Fault Probability Ratio as a function of number of minterms for n=8 at defect rates of 5%, 10% and 15%



Defect Rate (P_{S-ON}) Figure 5. Fault Probability Ratio as a function of defect rate for a range of fan-in (n) when $m=2^{n-1}$

Figure 5 shows that although the FPR rises with the defect rate, the rate of increase is more marked for circuits with higher fan-in (n). Moreover, for a particular defect rate, the FPR is greater for circuits with higher fan-in. For instance, at a defect rate of 12%, a tile with a fan-in of 6 had an FPR of 4.5 whereas a tile with a fan-in of 10 had

an FPR of 12.7. This can be attributed largely to the increase in the number of minterms and thus increased chances of any one of them carrying a faulty '0' and switching a 2nd stage NAND gate transistor incorrectly off.

4.3 Biased Voting in General Case

In this section, we compare biased voting schemes to conventional majority voting schemes for general case without regard to any specific fabric. Figure 6 shows a TMR and a biased voting configuration. The key difference between the two is the voting decision: in the first case (TMR) a majority voter is used to vote on three structurally non-redundant input modules whereas in the second, a voter biased towards a single logic '0' is used. The notation used for this biased voter is V01/3 (voter biased towards logic '0' that will produce logic '0' even if only 1 out of the 3 inputs is '0'). A biased voter has a lower (higher) tolerance for the less (more) prevalent fault type. For instance, the V01/3 biased voter will not be able to tolerate any faulty '0's but can tolerate up to two faulty '1's.

Our objective is to find out whether the use of a biased voter instead of a conventional majority voter can enhance the expected yield of the voting configuration.



Figure 6. Voting configurations: (a) TMR majority voting, (b) $V_0^{1/3}$ *biased voting*
Yield is the probability that the circuit will produce a correct output in the presence of manufacturing defects. The two voting configurations (shown in Figure 6) use the same three non-redundant nano-modules. Thus, the difference in the probability of producing a correct output by the two circuits will be due to the different voting schemes. Such an accurate comparison between the two voting schemes and their corresponding voter designs can be achieved by using the Signal Reliability metric [43],[44].

To derive the probabilities of producing correct '0' and '1' outputs by the two voting configurations we use the notations shown in Table 2.

Symbol	Description				
V ^C	Correct output from the voting scheme if it were to take inputs from defect-free input modules				
V^{A}	Actual output from the voting scheme taking inputs from defective input modules				
$P_{i/j}^{out} = P_{i/j}$	Conditional probability of getting a input module output of <i>i</i> given that the correct module output should have been <i>j</i> (<i>i</i> , <i>j</i> =0,1). Clearly, $P_{0/j}^{out} + P_{1/j}^{out} = 1$, for j=0,1.				
$P_{i/j}^{v}$	Conditional probability of getting a voter output of <i>i</i> given that the correct voter output should have been j ($i,j=0,1$) based on inputs to the voter. <i>i</i> may differ from j only due to defects within the voter circuit				
$P_{i/j}^{TMR}$	Conditional probability of getting a TMR majority voting scheme output of <i>i</i> given that the correct output should have been <i>j</i> (<i>i</i> , <i>j</i> =0,1) based on inputs to the input modules. <i>i</i> may differ from <i>j</i> due to defects within the voter circuit and/or defects in the input modules				
$P_{i/j}^{V01/3}$	Conditional probability of getting a $V_0^{1/3}$ biased voting scheme output of <i>i</i> given that the correct output should have been <i>j</i> (<i>i</i> , <i>j</i> =0,1) based on inputs to the input modules. <i>i</i> may differ from <i>j</i> due to defects within the voter circuit and/or defects in the input modules				

Table 2. Notations

As a preliminary study, faults in voters are not considered in this section to simplify the resulting expressions. However, section 4.5 discusses the changes in results when voters are prone to defects. Thus in this section, $P_{0/1}^v = P_{1/0}^v = 0$.

Assuming ideal voter circuits, a correct '0' ('1') will be produced by a TMR if all three inputs to the voter (from the modules) are correct '0's ('1's) or if any two of the inputs are correct '0's ('1's). The probabilities that a TMR generates correct '0's are,

$$P_{0/0}^{TMR} = Prob\{V_{TMR}^{A} = 0 \mid V_{TMR}^{C} = 0\}$$
(4.2)

$$P_{0/0}^{TMR} = (1 - P_{1/0})^3 + 3P_{1/0}(1 - P_{1/0})^2$$
(4.3)

$$P_{0/0}^{TMR} = (P_{0/0})^3 + 3P_{1/0}(P_{0/0})^2$$
(4.4)

The probabilities that a TMR generates correct '1's are,

$$P_{1/1}^{TMR} = Prob\{V_{TMR}^{A} = 1 \mid V_{TMR}^{C} = 1\}$$
(4.5)

$$P_{1/1}^{TMR} = (1 - P_{0/1})^3 + 3P_{0/1}(1 - P_{0/1})^2$$
(4.6)

$$P_{1/1}^{TMR} = (P_{1/1})^3 + 3P_{0/1}(P_{1/1})^2$$
(4.7)

Similarly, probability expressions for the V01/3 biased voter can be derived,

$$P_{0/0}^{V01/3} = Prob\{V_{V01/3}^{A} = 0 \mid V_{V01/3}^{C} = 0\}$$
(4.8)

$$P_{0/0}^{V01/3} = (1 - P_{1/0})^3 + 3P_{1/0}(1 - P_{1/0})^2 + 3P_{1/0}^2(1 - P_{1/0})$$
(4.9)

$$P_{0/0}^{V01/3} = (P_{0/0})^3 + 3P_{1/0}(P_{0/0})^2 + 3P_{1/0}{}^2P_{0/0}$$
(4.10)

$$P_{1/1}^{V01/3} = (1 - P_{0/1})^3 = P_{1/1}^{3}$$
(4.11)

The values of $P_{i/j}^{TMR}$ and $P_{i/j}^{V01/3}$ are dependent on the defect rate, defect model and the circuit and logic styles of the nano-fabric considered.

The signal reliability of the voting scheme's output can be expressed as:

$$SR^{TMR} = [P_{0/0}^{TMR} \times P(V^{C} = 0)] + [P_{1/1}^{TMR} \times P(V^{C} = 1)]$$
(4.12)

$$SR^{V01/3} = [P_{0/0}^{V01/3} \times P(V^{C} = 0)] + [P_{1/1}^{V01/3} \times P(V^{C} = 1)]$$
(4.13)

The signal reliability expressions for the TMR and V01/3 can be obtained by substituting Equations 4.4 & 4.7 and 4.10 & 4.11 into Equation 4.12 and 4.13 respectively. Using a similar approach, the signal reliability for other majority and biased voting configurations can be found. It is evident from Equations 4.12 and 4.13 that the signal reliability not only depends on the probability of faulty '1's and '0's produced by the input modules but also on the number of input patterns that should correctly generate '0's P(V^C = 0) and those that should correctly generate '1's P(V^C = 1). Figure 7 shows the signal reliabilities for four different voting schemes over a range of Fault Probability Ratios (FPR), where the FPR is the ratio between the probability of a faulty '1' to the probability of a faulty '0' as defined in Equation 4.1. The $V_0^{2/5}$ biased voting scheme is biased towards 2 '0's out of the 5 inputs and the 5MR is a majority voting scheme voting on 5 redundant inputs. Figure 7 (a) and (b) show the signal reliability for P_{0/1} values of 0.05 and 0.1, for a symmetric function where P(V^C = 0) = P(V^C = 1) = 0.5. However,

it is also possible for a logic function to be skewed towards '1' or '0' in terms of correct outputs. Figure 7 (c) and (d) show the signal reliabilities for the cases where $P(V^{C} = 1) = 0.75$ and $P(V^{C} = 0) = 0.25$, respectively.

As can be seen from Figure 7 (a) and (b), the biased voting schemes $(V_0^{1/3} \text{ and } V_0^{2/5})$ have a higher signal reliability compared to their corresponding majority voting schemes (TMR and 5MR, respectively) with increasing FPR. As the probability of a faulty '1' rises compared to that of a faulty '0', the voters biased to logic '0' will be more likely to produce a correct output. For example, in Figure 7 (b) where $P_{0/1} = 0.1$ with $P(V^C = 0) = P(V^C = 1) = 0.5$, the $V_0^{2/5}$ scheme has 20% higher signal reliability than the 5MR majority scheme while the $V_0^{1/3}$ biased scheme has a 16% higher signal reliability than the TMR at FPR = 7.

The $V_0^{2/5}$ scheme exhibits higher signal reliability than 5MR for FPR ≥ 3 , whereas for the $V_0^{1/3}$ scheme the benefits over TMR are achieved only for FPR > 4. This is due to the comparatively poor resilience of the latter scheme to faulty '0's, since a single faulty '0' causes diminished reliability in that scheme. In this case, the probability of generating correct '1's at the output of the voter is the dominating factor. On the other hand, for FPR > 7, the faulty '1' probability is much higher and the requirement for 2-out-of-5 '0's for the $V_0^{2/5}$ voter is harder to achieve. However in the case of the $V_0^{1/3}$ voter, a single correct '0' is sufficient to ensure a correct final output implying that the overall reliability is higher. The probability of generating correct '0's is the dominating factor in this scenario.



Figure 7. Signal reliability as a function of Fault Probability Ratio (FPR) for (a) $P_{0/1} = 0.05$, $P(V^{c}=0) = 0.5$ and (b) $P_{0/1} = 0.1$, $P(V^{c}=0) = 0.5$ (c) $P_{0/1} = 0.1$, $P(V^{c}=0) = 0.75$ and (d) $P_{0/1} = 0.1$, $P(V^{c}=0) = 0.25$

In situations where $P(V^{C} = 0) > P(V^{C} = 1)$ such as the case shown in Figure 7, the frequency of occurrence of faulty '1's (0's) is further increased (reduced). At an FPR of 7, a 57% increase in reliability is attained by the $V_0^{2/5}$ scheme over the 5MR while a 66% increase in reliability is demonstrated by the $V_0^{1/3}$ scheme over the TMR. Figure 7 (d) shows the opposite case where $P(V^{C} = 0) < P(V^{C} = 1)$. Here the frequency of occurrence of faulty '1's is reduced while that of faulty '0's is increased. The $V_0^{2/5}$ scheme is still able to show a higher reliability from an FPR value of 4 up to 8. On the other hand, the $V_0^{1/3}$ scheme shows low reliability because of its inability to tolerate even a single faulty '0'. This suggests that a higher reliability maybe attained by switching the bias of the voters from logic '0' to '1'. However, in nanoscale fabrics that implement two level logic styles, for functions where $P(V^C = 0) < P(V^C = 1)$, the complementary rather than the true output is likely to be implemented for reduced area. This again takes us back to the reverse case depicted in Figure 7 where the biased voters demonstrate significantly higher signal reliabilities.

To summarize, our analysis of biased and majority voting schemes has shown that as the ratio of faulty '1's to faulty '0's increases, biased voting configurations exhibit up to 66% higher signal reliability.

4.4 Biased Voting in NASICs

Section 4.2 showed that the rate of occurrence of faulty '1's is expected to be higher than that of faulty '0's in NASICs and section 4.3 demonstrated the improvement in signal reliability that can be gained by applying biased voters to fabrics that have unequal fault rates. It follows that applying biased voting schemes to NASICs will provide improved fault tolerance compared to conventional majority voters. In this section, we apply biased voting schemes to NASICs and compare them to conventional majority voting schemes.

In this section, the probability expressions for faulty '1's and faulty '0's derived in Chapter 3, are used to compare the signal reliabilities for circuits implemented with biased and majority voting schemes. Figure 8 shows the reliabilities for n-input NASIC circuits for the voting schemes introduced in the previous section. As we have previously observed for a given defect rate, there is a minimum fan-in (n) before biased voting schemes show a higher reliability over majority voting schemes. Moreover, this required minimum value of fan-in decreases with increasing defect rates. For example, at a defect rate of 5%, the $V_0^{1/3}$ biased voting scheme has a greater reliability than the TMR scheme only for NASIC circuits with n = 9 or greater. However, at a defect rate of 15%, the $V_0^{1/3}$ scheme has a higher reliability even for circuits that have a fan-in of 6. This suggests that at the high defect rates that nanoscale fabrics are subject to, using biased voting schemes will provide a greater reliability even for circuits with low fan-in.

For a particular defect rate, the biased voting schemes give a higher reliability as the circuit fan-in increases due to the increasing FPR discussed in the previous section. For instance at 15% defect rate for tiles with fan-in of 10, the $V_0^{1/3}$ biased scheme has a 27% greater reliability compared to the TMR scheme while the $V_0^{2/5}$ biased scheme has 24% higher reliability than the 5MR scheme. At a defect rate of 10%, the $V_0^{2/5}$ exhibits 20% greater signal reliability and the $V_0^{1/3}$ exhibits 17% greater signal reliability for circuits with fan-in of 10. The reliability advantage will be even greater when $P(V^C = 0) \neq P(V^C = 1)$ as shown in section 4.3. This implies that biased voters could be employed at key architectural points in the design, specifically at the outputs of high fanin stages, for greater yield, while carefully managing yield-area tradeoffs.



 \leftarrow TMR \leftarrow 5MR \leftarrow V01/3 \rightarrow V02/5

Figure 8. Signal Reliability comparison for biased and majority voting schemes at defect rates of: (a) 5%, (b) 10%, and (c) 15%

The biased voting schemes may also be more area efficient than majority voters or, at the worst case, consume the same area when implemented with 2-level logic such as used in the NASIC fabric. This is because the number of minterms for a biased voter will be less than or equal to the number of minterms for a majority voter with the same number of inputs. This implies that the effective yield (i.e., yield divided by the area increase factor) for biased voting schemes may be higher.

4.5 Voter Reliability

In the analysis done in the previous sections of this chapter, voters were assumed to be perfect. However, since the voters have to be implemented in the same nanoscale fabric, such as the $V_0^{2/4}$ biased voter shown in Figure 9 implemented in the NASICs fabric, they are also susceptible to manufacturing defects. Thus in a realistic scenario voter faults have to be taken into account. In this section, we analyze and compare signal reliabilities in voting schemes with voter defects. Notations that have been used in this section are described in Table 2.

A correct '0' ('1') will be produced by a TMR majority voting scheme if a majority of the input modules produce correct '0's ('1's) and the voter circuit itself produces a correct '0' ('1') or if a majority of the input modules produce incorrect '1's



Figure 9. $V_0^{2/4}$ biased voter in NASICs.

('0's) and the voter circuit produces an incorrect '0' ('1'). The probabilities that this majority voting scheme generates correct '0's and '1's are,

$$P_{0/0}^{\text{TMR}} = P_{0/0}^{\text{V}}[(P_{0/0})^3 + 3P_{1/0}(P_{0/0})^2] + P_{0/1}^{\text{V}}[(P_{1/0})^3 + 3P_{0/0}(P_{1/0})^2]$$
$$P_{1/1}^{\text{TMR}} = P_{1/1}^{\text{V}}[(P_{1/1})^3 + 3P_{0/1}(P_{1/1})^2] + P_{1/0}^{\text{V}}[(P_{0/1})^3 + 3P_{1/1}(P_{0/1})^2]$$

Similarly, probability expressions for the $V_0^{1/3}$ biased voter can be derived,

$$P_{0/0}^{V01/3} = P_{0/0}^{V} [1 - (P_{1/0})^3] + P_{0/1}^{V} [(P_{1/0})^3]$$
$$P_{1/1}^{V01/3} = P_{1/1}^{V} [(P_{1/1})^3] + P_{1/0}^{V} [1 - (P_{1/1})^3]$$

The signal reliability of the TMR majority voting scheme and the $V_0^{1/3}$ biased voting scheme can be expressed as follows where i = 0 or 1:

$$SR^{TMR} = [P_{0/0}^{TMR} \times P_{i/0}^{TMR}] + [P_{1/1}^{TMR} \times P_{i/1}^{TMR}]$$
$$SR^{V01/3} = [P_{0/0}^{V01/3} \times P_{i/0}^{V01/3}] + [P_{1/1}^{V01/3} \times P_{i/1}^{V01/3}]$$

The signal reliability expressions for the TMR and $V_0^{1/3}$ can be obtained by substituting Equations 4.14 & 4.15 and 4.16 & 4.17 into Equations 4.18 and 4.19 respectively. The values for $P_{i/j} = P_{i/j}^{out}$ can be calculated based on the equations derived in Chapter 3.5 for NASICs.

The values for $P_{0/0}^V$ for the TMR and the $V_0^{1/3}$ biased voted can also be calculated in a similar manner and is determined by the fan-in and number of minterms in the voter circuit. For instance, a structurally non-redundant TMR voter would have a fan-in of 3 and 3 minterms and a $V_0^{1/3}$ would have the same fan-in but only one minterm since it will produce a '1' only when all three inputs to it are '1's.

Figure 10 shows the signal reliabilities for the same voting schemes that were considered in Figure 8 but with consideration of voter defects. As expected, the signal reliabilities overall decrease since the voters are now defective. It is interesting to note,



Figure 10. Signal Reliability comparison for biased and majority voting schemes after considering voter defects at defect rates of: (a) 5%, (b) 10%, and (c) 15%

however, that the biased schemes show an advantage over majority schemes much earlier on now. For instance at defect rate of 10% in Figure 10 (a) $V_0^{2/5}$ biased scheme now has greater signal reliability for circuits of any fan-in and the $V_0^{1/3}$ voter overtakes the TMR for circuits with fan-in greater than 4 as opposed to 6. This is because the biased voter circuits require fewer minterms than majority voter circuits and hence the number of transistors in the biased voter circuits is fewer. Thus they have a lower probability of being defective compared to majority voters which have larger circuits. On the same note, biased voters are expected to take lesser area than majority voters and thus provide greater signal reliability per unit area.

4.6 Summary

In this chapter we have shown that for structurally non-redundant NASIC tiles with non-faulty inputs, the probability of output faulty '1's can be as much as 12 times greater than the probability of output faulty '0's for a tile with a fan-in of 10. It was shown that this ratio can be even greater for circuits with a higher fan-in and at greater defect rates. We have shown that the application of biased voting schemes as compared to conventional majority voting schemes provided improved signal reliability in NASICs due to the inequality in fault rates. The biased voting schemes were shown to have as much as 27% greater signal reliability compared to conventional majority voting schemes. This advantage over majority voting schemes was shown to be retained when voter defects were considered.

CHAPTER 5

STRUCTURAL REDUNDANCY IN CASCADED NASICS

5.1 Introduction

In the previous chapters we utilized analytical models to analyze fault characteristics in non-cascaded and structurally non-redundant NASIC tiles with ideal inputs. To build larger systems, however, NASIC tiles will have to be cascaded. Hence, evaluation of the fault characteristics and improvement of the signal reliability through cascaded NASIC tiles is necessary. Furthermore, modular redundancy alone will not be sufficient to tolerate the high defect rates characteristic of nanoscale fabrics since multiple input modules may be faulty. Thus, structural redundancy needs to be incorporated in the tiles. In this chapter, we will study the propagation of faulty '1's and faulty '0's through cascades of structurally redundant NASIC tiles and their effect on the overall Signal Reliability. In our analysis, we demonstrate that the Signal Reliability through cascades of tiles with Regular Structural Redundancy rapidly drops after a few cascades due to an escalation of the probability of output faulty '0's. Having identified that tiles with Regular Structural Redundancy are intolerant of input faulty '0's, we propose and introduce a novel structural redundancy scheme called the Staggered Structural Redundancy (SSR) that offers improved resilience against input faulty '0's. We show that cascades of tiles with Staggered Structural Redundancy maintain high signal reliability for larger levels of cascades than were possible using Regular Structural Redundancy.



Figure 11: Regular Structural Redundancy schemes for a NASIC tile implementing a 2input XOR logic function: (a) NASIC tile with no structural redundancy incorporated (b) NASIC tile with 2-way RSR

5.2 Regular Structural Redundancy

Figure 11 shows a 2-input XOR NASIC tile without structural redundancy and its implementation with 2-way and 3-way Regular Structural Redundancy. In the 2-way RSR shown in Figure 11 (b), each nanowire has two Redundant Units and there are four such Redundant Units in total in the first stage of the tile.

5.2.1 Fault Propagation along cascade

In this section we analyze the propagation of faulty '1' and faulty '0' through a cascade of NASIC tiles with Regular Structural Redundancy. Depending on the specific NASIC system, tiles may be cascaded differently. For instance, output signals from a tile may be input to multiple tiles at different levels of the cascade. We analyze the propagation of faults through one example of the cascade structure shown in Figure 12 (another example of Ripple Carry Adder is shown in Section 5.4). In this cascade



Figure 12: Cascade structure used to analyze propagation of faults

structure, there are n levels of cascade. Each tile in the cascade is assumed to have the same number of inputs and minterms. All the inputs of a tile in the cascade are assumed to have the same probability of being faulty and this is equal to the probability of output faults from the preceding tile in the cascade.

In order to analyze the propagation of faults through the cascade shown in Figure 12, we have used the analytical model derived in Chapter 3. The model determined the probability of output faulty '1' and '0' in terms of the probability of defects within the tile and the probability of input faults from the preceding tile. This model is used to calculate the probability of output faults after each level of the cascade using the values from the preceding tile as its probability of input faults. Inputs to the tile in Level 1 of the cascade are assumed to be fault-free.

Figure 13(a) shows the probability of output faults and the Signal Reliability (SR) at the output of each cascade level along a cascade of NASIC tiles with no structural redundancy (1w-RSR) and with tile specifications of n=4 and m=8 at a defect rate of 10%. The Signal Reliability (SR) is the probability of a correct output as was defined in Chapter 4. In our analysis we assume that output '0' and '1' are equally likely at the output of an ideal NASIC tile. Hence, the Signal Reliability is $SR = 0.5 \times (P_{1/0} + P_{0/1})$



Figure 13: Probability of output faulty '0' and faulty '1' and Signal Reliability of cascaded tiles with n=4 and m=8 at 10% stuck-on probability for (a) tiles with no structural redundancy (b) with 1w-RSR and (c) with 2w-RSR

where $P_{1/0}$ and $P_{0/1}$ is the probability of output faulty '1' and faulty '0' respectively. Figure 13(a) shows that the probability of output faulty '1' is 0.24 after the first cascade level and the probability of output faulty '0' is 0.08. The Fault Probability Ratio is thus greater than 1 and this is consistent with our analysis in Chapter 4.1. It can be seen from Figure 13(a) that the probability of faulty '1' increases along the cascade suppressing the probability of faulty '0'. This is because as the rate of input faulty '1's increases, transistors in the first NAND stage of a NASIC tile (such as the shown in Figure 11(a)) are incorrectly switched-on. This increases the rate of minterm faulty '0's causing transistors in the 2nd NAND stage to be incorrectly switched-off thus causing output faulty '1's. For the same reason, the rate of output faulty '0's are reduced. This can also be explained using the analytical models derived in Chapter 3. As $P_{1/0}^{inp}$ increases in Equation 3.8, $P_{0N/OFF}^{T1}$ increases and this causes $P_{0/1}^{min}$ to also increase in Equation 3.7. With that $P_{0N/ON}^{T2}$ reduces in Equation 3.4 finally causing $P_{1/0}^{out}$ in Equation 3.5 to increase and $P_{0/1}^{out}$ to decrease in Equation 3.12.

It can be seen from Figure 13(a) that after 3 levels of cascade, the probability of faulty '1' reaches 1 and the probability of faulty '0' reaches 0. This means that the output of the third cascade level will be stuck at logic '1'. The Signal Reliability is 0.5 because about half the time logic '1' is the correct value. Thus, clearly NASIC tiles cannot be cascaded without structural or modular redundancy since the faults will accumulate along the cascade.

Figure 13(b) shows the results for a cascade of the same tile specifications but implemented with 2-way Regular Structural Redundancy (2w-RSR). It can be seen that the Signal Reliability in this case drops to 0.5 much later in the cascade, i.e. after 6 levels, compared to the previous case. This is because every transistor has a redundant copy thus providing tolerance against stuck-on defects. However, it is interesting to observe that the reason for the drop in Signal Reliability in this cascade, is the escalation of probability of output faulty '0's instead of faulty '1's, unlike the previous case. Although the probability of output faulty '0' starts off with a lower value compared to the probability of output faulty '1' after level 1 of the cascade, it quickly dominates and saturates to 1 after 6 levels of cascade. This leads us to conclude that tiles with Regular Structural Redundancy have reduced resilience against input faulty '0's and improved resilience against input faulty '1's compared to structurally non-redundant tiles. Hence, if the resilience against input faulty '0's from preceding tiles could be improved alongside tolerating stuck-on defects within the structurally redundant tile, then Signal Reliability can be further improved, enabling larger cascades.

We have identified two factors that cause diminished resilience against input faulty '0' in tiles with Regular Structural Redundancy. Firstly, for tiles with RSR the probability of the minterm being faulty '1' is greater. This is due to the fact that a single incorrectly switched-off transistor in a horizontal nanowire (induced by an input faulty



Figure 14: 3w-RSR tile demonstrating diminished resilience to input faulty '0'

'0') is sufficient to cause the minterm to be faulty '1' and furthermore, the number of transistors in the horizontal nanowire is multiplied due to redundancy. For instance, a 3w-RSR tile has three times the number of transistors in each horizontal nanowire compared to a non-redundant tile. Secondly, logical dependencies exist between redundant copies of horizontal nanowires because they are gated by the same input signals. Thus, a faulty '0' input may cause all redundant copies of a horizontal nanowire to produce faulty minterm '1's leading to faulty output '0' at the vertical nanowire. Figure 14 shows the 3w-RSR tile from Figure 11(c) this time with inputs. The correct inputs to the tile (shown in green) are a = 1 and b = 0 which in an ideal scenario would produce an output y = 1. However if any of the input '1's become a faulty '0', such as \overline{b}' shown in the figure, then the relevant transistors in all three horizontal nanowires are incorrectly switched-off. In the example shown in Figure 14, this causes all the three transistors shown in yellow in the vertical nanowire to be incorrectly switched-on resulting in a faulty '0' at the output. In section 5.3 we propose a novel structural redundancy scheme that circumvents this problem to improve resilience against input faulty '0's while tolerating stuck-on defects.

5.2.2 Resilience to Input Faults

In this section, we study the resilience of the 3w-RSR scheme against input faulty '0's. Figure 15(a) shows a 4-input NASIC tile with ideal input '1's and faulty '0's with a probability of $P_{0/1}^{inp}$. Figure 15(b) shows the probability of output faulty '1's and '0's as the probability of input faulty '0' varies from 0 to 0.5. The results were obtained using the analytical fault models from Chapter 3. When $P_{0/1}^{inp} = 0$, the probability of output faults for 1w-RSR is the greatest because it does not have any redundancy against the stuck-on defects. The probability of output faulty '1's is greater than faulty '0's due to the reasons stated in Chapter 4. As the probability of input faulty '0's increases, probability of output faulty '0's for the 3w-RSR becomes greater than the 2w-RSR and the 1w-RSR.



Figure 15: Resilience to input faulty '0' by non-redundant tile, 2w-RSR and 3w-RSR for n=4, m=8. (a) 4-input tile with 10% defect rate and with faulty input '0's. (b) Probability of output faults as a function of probability of input faulty '0's

This is because the 3w-RSR has three times the number of input signals due to redundancy compared to the non-redundant signal coupled with the fact that a single input faulty '0' may cause an output faulty '0' as explained in the preceding section.

5.3 Staggered Structural Redundancy

5.3.1 Introduction

Having identified the rapid escalation of faulty '0's along the cascade of tiles with Regular Structural Redundancy, the goal in designing the new Staggered Structural Redundancy is to increase tolerance against input faulty '0's. This will delay or even avoid the 'avalanche' of faulty '0's making larger cascades feasible. In Staggered Structural Redundancy (SSR), we propose to selectively remove certain redundant units from the corresponding Regular Structural Redundancy such that each Redundant Input Group (RIG) does not gate transistors on all 3 redundant copies of each horizontal nanowire. Figure 16 shows one example of the Staggered Structural Redundancy, the (3,2)-SSR. It is so named because it has 3 RIGs and each horizontal nanowire is 2-way redundant. The (3,2)-SSR is more resilient against input faulty '0' compared to the 3w-RSR in Figure 14. For instance, if the same input, $\overline{b'}$, is faulty '0', transistor T' in the vertically nanowire is not affected and is correctly switched-off to produce the correct output of '1'. Thus, assuming no defects, the (3,2)-SSR tile is guaranteed to tolerate at least a single input faulty '0' whereas the 3w-RSR in Figure 14 was unable to tolerate even a single input faulty '0'. On the other hand, the (3,2)-SSR has reduced tolerance to stuck-on defects compared to compared to the 3w-RSR since each horizontal nanowire is

now 2way redundant instead of 3way. Thus, at very high defect rates the (3,2)-SSR will show reduced advantage over the 3w-RSR and this is discussed in Section 5.3.5.

5.3.2 Criteria and Notation

In order to define criteria for the Staggered Structural Redundancy schemes, we first define matrix notations for the schemes that relates to the presence or absence of Redundant Units in the first NAND stage of the tile. In the m \times m square matrix, 1 indicates the presence of Redundant Units of transistors and 0 indicates absence. Thus, the matrix notation for the 3w-RSR shown in Figure 14 will be 3×3 square matrix with all elements as 1.

The criteria that needs to be fulfilled by the Staggered Structural Redundancy scheme to achieve improved resilience to input faulty '0's without considerable loss of tolerance of stuck-on defects are as follows:

• At least a single 0 and at least a single 1 in each column of the matrix notation for



Figure 16. (3,2) staggered structural redundancy of 2-input XOR gate in NASICs

the staggered structural redundancy.

• An equal number of 1s and an equal number of 0s in each column of the matrix notation.

The above two criteria ensure that at least a single input faulty '0' is tolerated by a tile assuming the tile itself is defect-free. Furthermore, it ensures that all horizontal nanowires are equally redundant. The presence of a lesser redundant nanowire in the tile could make the whole tile susceptible to stuck-on defects or input faulty '1's. Note that for each SSR scheme there may be more than one possible matrix notation. However, in terms of logic implementation and input fault tolerance, the different matrix notations will have the same effect. Table 3 shows the matrix notations of 3w-RSR and two possible SSR schemes with the same number of Redundant Input Groups, the (3,2)-SSR and the (3,1)-SSR.

Structural Redundancy Type	Possible Matrix Notation		
3w-RSR:	$\begin{bmatrix} 1 & 1 & 1 \\ 1 & 1 & 1 \\ 1 & 1 & 1 \end{bmatrix}$		
(3,2)-SSR:	$\begin{bmatrix} 0 & 1 & 1 \\ 1 & 0 & 1 \\ 1 & 1 & 0 \end{bmatrix}$		
(3,1)-SSR	$\begin{bmatrix} 0 & 0 & 1 \\ 1 & 0 & 0 \\ 0 & 1 & 0 \end{bmatrix}$		

Table 3: Matrix notation of structural redundancy schemes

The (3,1)-SSR scheme has three Redundant Input Groups and each horizontal nanowire in the tile is 1 way redundant. In other words, there are no redundant transistors

presents. Thus, although the tile would be more tolerant of input faulty '0's, it is more susceptible to stuck-on defects and input faulty '1's compared to the (3,2)-SSR.

5.3.3 Simulation Framework

The analytical models derived in Chapter 3 did not take into account logical dependencies between horizontal nanowires in order to keep the models tractable. Thus to achieve more accurate comparisons of the fault tolerance of the Staggered Structural Redundancy and the Regular Structural Redundancy schemes, Monte Carlo simulations have been used. The simulator first creates a faulty tile by mapping stuck-on defects on the ideal tile based on the probability of stuck-on defects. It then generates an input pattern from a set of all possible input patterns with uniform distribution. Faulty input patterns are created by flipping the bits based on the probability of input faulty '1' and faulty '0'. The ideal inputs are fed into the defect-free tile and faulty inputs are fed into the faulty tile and the outputs are compared. Based on the comparison results of 100,000 such iterations the probability of output faults and the Signal Reliability are computed.

5.3.4 Resilience to Input Faults

Figure 17 compares the resilience to input faulty '0' offered by the 3way Regular Structural Redundancy and the (3,2) Staggered Structural Redundancy schemes. Figure 17(a) shows the NASIC tile simulated for this purpose which is a 4-input tile implementing the XOR function. Simulations were carried out for the 3w-RSR and the (3,2)-RSR schemes with a 10% defect rate, probability of input faulty '1' as 0 and varying the probability of input faulty '0' from 0 to 0.4. Figure 17(b) shows the probability of output faulty '1' and the probability of output faulty '0' for the two schemes. The probability of output faulty '0' increases and the probability of output faulty '1' decreases or stays at 0 for both schemes with increasing probability of input faulty '0'. This is because input faulty '0's cause minterms to be faulty '1's which in turn causes output faulty '0's. Also, the presence of input faulty '0's may mask the effect of stuck-on transistors in the horizontal nanowires and avoid minterm faulty '0's, thus reducing the probability of output faulty '1's. This can be explained using the analytical



Figure 17: Resilience to input faulty '0' by 3w-RSR and (3,2)-SSR for n=4, m=8. (a) 4-input XOR tile with 10% defect rate and with faulty input '0's. (b) Probability of output faults for the two schemes (c) Signal Reliability for the two schemes

models just as was done for input faulty '1's in Section 5.3.1.

From Figure 17(b) it can be seen that the probability of output faulty '0' in the (3,2)-SSR is lower than that of the 3w-RSR. For instance, at a probability of input faulty '0' of 0.1, the probability of output faulty '0' for the 3w-RSR is 0.62 and that for the (3,2)-SSR is 0.27 representing a 56% improvement over the 3w-RSR. This is because in the (3,2)-SSR scheme, a single faulty '0' does not affect all the redundant horizontal nanowires due to the absence of Redundant Units as shown in Figure 16. As a result, at least one transistor in the corresponding vertical nanowire will be correctly switched-off (if it is not defective) and prevent an output faulty '0'. On the other hand, in the 3w-RSR, a single faulty '0' may affect cause all transistors in the corresponding vertical nanowire to switch-on, leading to an output faulty '0'. It can also be seen that the probability of output faulty '1' of the (3,2)-SSR is slightly higher compared to the 3w-RSR at low probability of input faulty '0's. This is because in the (3,2)-SSR each horizontal nanowire is 2way redundant whereas in the 3w-RSR they are 3way redundant. Hence (3,2)-SSR has reduced tolerance to stuck-on defects and to input faulty '1's which may cause the minterm to be faulty '0' leading to output faulty '1's. However, having identified the rapid escalation of faulty '0's along a cascade of 3w-RSR, the benefits of increased tolerance to input faulty '0' by the (3,2)-SSR is expected to outweigh the reduced tolerance to stuck-on defects and input faulty '1's at low enough defect rates along the cascade.

Figure 17(c) shows the overall Signal Reliability of the output of the two redundancy schemes. It can be seen that the Signal Reliability of the (3,2)-SSR is greater than that of the 3w-RSR at a particular probability of input faulty '0'. For instance, at a

probability of input faulty '0' of 0.1, the Signal Reliability of the (3,2)-SSR is 0.86 whereas that of the 3w-RSR is 0.69 representing a 24% increase in Signal Reliability compared to the 3w-RSR. This increase in Signal Reliability is dominated by the reduced probability of output faulty '0' of the (3,2)-SSR.

5.3.5 Fault Propagation along cascade

Figure 18 shows the fault probabilities and Signal Reliability at each level of the cascade shown in Figure 12 for 2w-RSR, 3w-RSR and (3,2)-SSR. Each NASIC tile is simulated using the simulation framework described in Section 5.3.3 with the output fault probabilities of one tile being the input fault probabilities for the tile in the next cascade level.

Figure 18(a) shows the probability of output faulty '1' and '0' for the three schemes at 10% defect rate. It can be seen that the probability of output faulty '0' for the 2w-RSR and 3w-RSR rises rapidly after the 2nd and 3rd cascade level, respectively, whereas the escalation of output faulty '0's in the (3,2)-SSR is delayed until the 17th level thus marking a significant improvement. On the other hand, the probability of output faulty '1' is slightly higher for the (3,2)-SSR due to its reduced tolerance to input faulty '1's and susceptibility to stuck-on defects compared to corresponding RSR schemes. Figure 18(b) shows the overall Signal Reliability for the three schemes. The trends are seen to be strongly dominated by the probability of output faulty '0's as expected. The Signal Reliability for the 3w-RSR is better compared to the (3,2)-SSR before the 3rd cascade level, because during those initial cascades, the Signal Reliability is dominated by the output faulty '1's since the probability of output '0's is still low.



Figure 18:(a)Probability of output faults and (b) Signal Reliability along cascade of tiles with n=4 and m=8 at 10% defect rate

Figure 19(a) and (b) shows the Signal Reliabilities for the three schemes at 5% and 10% defect rates, respectively.

It can be seen from Figure 19(a) that the Signal Reliability of the (3,2)-SSR is nearly 1 for at least 15 cascades whereas the Signal Reliability of the 2w-RSR and the



Figure 19: Signal Reliability along cascade of 2w-RSR, 3w-RSR and (3,2)-SSR for tiles with n=4, m=8 and at (a) 5% defect rate and (b) 15% defect rate

3w-RSR drop to 0.5 after 6 and 7 cascade levels, respectively. In fact, the Signal Reliability of the (3,2)-SSR have been found to be over 0.98 for more than 100 cascade levels. This is because the output faulty '1's and '0's that are produced by a (3,2)-SSR tile at 5% defect rate, is effectively tolerated by the next tile in the cascade avoiding

magnification of fault probabilities. As a result an 'avalanche' of faults is avoided keeping the Signal Reliability above 0.98 for a significantly large number of cascade levels.

Table 4 lists the cascade levels at which the Signal Reliability drops to 0.5 for the 2w-RSR, 3w-RSR and (3,2)-SSR schemes at different defect rates up to 15%. For defect rates below 8%, the Signal Reliability of the (3,2)-SSR scheme does not drop below 0.98 for at least 100 levels of cascade. The maximum cascade level with Signal Reliability above 0.5 for the (3,2)-SSR starts decreasing rapidly for increasing defect rates above 8%. This is because more transistors in the 2nd NAND stage of the tile are stuck-on leading to output faulty '0's thus causing an 'avalanche' much earlier on in the cascade. It is interesting to note that at 15% defect rate, the gradual fall of the Signal Reliability of the (3,2)-SSR shown in Figure 19(b) is due to a gradual rise in output faulty '1's. This occurs because each horizontal nanowire in the (3,2)-SSR is 2way redundant thus producing more output faulty '1's. The increased number of output faulty '1's mask output faulty '0's and eventually dominate.

	Defect Rates					
	3%	5%	8%	10%	15%	
2w-RSR	6	6	5	4	4	
3w-RSR	7	7	6	5	5	
(3,2)-SSR	>100	>100	75	20	12	

Table 4: Cascade level after which Signal Reliability is 0.5

5.4 Example: Ripple Carry Adder

In this section, we compare the degradation of the Signal Reliability of the carry signal through a 16-bit ripple carry adder implemented with 3w-RSR and (3,2)-SSR.



Figure 20: (a) 16-bit Ripple carry adder with ideal inputs and faulty carry signal. (b) Signal Reliability of carry signal at each cascade level for 3w-RSR and (3,2)-SSR at 10% defect rate (b) Signal Reliability of the (3,2)-SSR at 5%, 10%, 15% and 20% defect rates

Figure 20(a) shows the 16-bit ripple carry adder with ideal inputs and C_{in} to the first tile in the cascade. The reliability of the carry signal degrades through the cascade. Figure 20(b) shows the signal reliability of the carry signal at the output of each cascade level for the implementations, 3w-RSR and (3,2)-SSR, at 10% defect rate. The signal reliability of the carry signal in the 3w-RSR implementation drops to 0.5 after 11 cascades levels. However, the signal reliability of the carry out signal at the end of the 16 cascade levels in the (3,2)-SSR implementation is 0.97, marking a significant improvement over the 3w-RSR scheme. Figure 20(c) shows the signal reliability of the carry signal in the (3,2)-SSR implementation at 5%, 10%, 15% and 20% defect rates. The signal reliability of the carry signal after 16 levels of cascade are 0.99, 0.97, 0.94 and 0.64 respectively. The signal reliability is lower with higher defect rates because the probability of faulty '1's increases with increasing probability of stuck-on transistors as was shown in Chapter 3.

5.5 Example: Cascade of 2-bit Comparators

In this section, we compare the Signal Reliabilities of signals in RSR and SSR implementations through a cascade of 2-bit comparators whose output is the maximum of two 2-bit numbers. Such a cascade of 16 levels is shown in Figure 21(a). Figure 21(b) shows the Signal Reliability of each of the two bits propagating along the cascade for 3w-RSR and (3,2)-SSR implementations at 5% defect rate. It can be seen that the Signal Reliabilities in the 3w-RSR implementation drops to their minimum value after 7 levels of cascade. After 7 levels of cascade, the output signals are all '0's. The minimum values are different for the two bits because the number of minterms is different and thus the probability that '0' is the correct output signal is different for the two bits. The Signal Reliabilities of the (3,2)-SSR scheme remain above 0.98 after 16 levels of cascade.





Figure 21: (a) Cascade of 2-bit comparators. (b) Signal Reliability along cascade of 3w-RSR and (3,2)-SSR implementations at 5%, (b) 10% and (c) 15% defect rate

Figure 21(b) shows the Signal Reliabilities at 10% defect rate. It can be seen that the degradation in Signal Reliability for the (3,2)-SSR scheme is greater in this case. Figure 21(c) shows the comparisons at 15% defect rate. It can be seen that the Signal Reliabilities of the (3,2)-SSR schemes reaches their minimum values after 10 levels of cascade. This occurs due to an escalation and eventual saturation of faulty '1's. The minimum value is different from the 3w-RSR case because in the 3w-RSR a saturation of faulty '0's occur.

5.6 Sensitivity to Stuck-Off Defects

In our analysis so far, we considered only stuck-on defects. In this section we provide a sensitivity analysis of the Signal Reliability through cascade implementations with stuck-off defects. Figure 22 shows the Signal Reliability along the cascade shown in Figure 12 with 10% total defect rate among which PS-OFF is the stuck-off rate and PS-ON is the stuck-on rate. $P_{S-ON} = 0.1 - P_{S-OFF}$. Figure 22(a) shows the Signal Reliability in 3w-RSR implementation of the cascade. It can be seen that significant drop in Signal Reliability occurs with even 1% rate of stuck-off defects. This is because a single stuck-off defect is sufficient to cause output of nanowire to be stuck at '1'.

Figure 22(b) shows the Signal Reliability in a (3,2)-SSR implementation. It can be seen that the SSR scheme is also equally affected by stuck-off defects. This is because, regardless of faults in the input signals, local stuck-off defects will have the same impact on both schemes since a single stuck-off transistors is sufficient to cause permanent fault at output a nanowire.

5.7 Summary

An analysis of fault propagation along a cascade of tiles with Regular Structural Redundancy was conducted using detailed analytical fault models. It was found that a rapid escalation of the probability of output faulty '0's occurred along such a cascade and this was the dominant factor in the sharp drop in Signal Reliability after a few cascade levels. A new redundancy scheme called the Staggered Structural Redundancy was proposed whose design was motivated by the need to improve tolerance against input faulty '0's in order make larger cascades feasible. Simulation results showed that a single tile with (3,2) Staggered Structural Redundancy offered up to 56% more resilience against input faulty '0's compared to the 3way Regular Structural Redundancy at 10% defect rate. The (3,2)-SSR scheme showed significant improvements in the Signal Reliability along a cascade of tiles compared to the 2w-RSR and the 3w-RSR. For instance, at a defect rate of 5%, the (3,2)-SSR Signal Reliability was greater than 0.98 after at least 100 levels of cascade whereas the that of the 3w-RSR dropped to 0.5 after 7 levels of cascade.



Figure 22: Signal Reliability along cascade with 10% total defect rate and PS-OFF stuck-off rate for (a) 3w-RSR and (b) (3,2)-SSR
CHAPTER 6 CONCLUSION

In this thesis, we developed detailed analytical fault models of the NASIC fabric taking into account its defect scenarios and its logic and circuit styles. Using the model, we have shown that the probability of output faulty '1's and '0's are unequal for single NASIC tiles. Consequently, application of biased voting schemes was shown to have up to 27% improvement on the signal reliability compared to conventional majority voting schemes. In this thesis, we also provided a study of the propagation of faults through a cascade of structurally redundant NASIC tiles. We have shown that a rapid escalation of faulty '0's occur along the cascade of tiles with Regular Structural Redundancy eventually leading to decreased signal reliability. Motivated by this, we designed and introduced novel Staggered Structural Redundancy schemes that provide up to 56% improved tolerance to input faulty '0's. We demonstrate that in a cascade of NASIC tiles employing the Staggered Structural Redundancy, the signal reliability of outputs at each cascade level remained above 0.98 for over 100 levels of cascade at 5% defect rate.

BIBLIOGRAPHY

- [1] W. Lu and C. M. Lieber, "Semiconductor nanowires," *Journal of Physics D: Applied Physics*, (2006), vol. 39, no. 21, pp. R387–R406.
- [2] Y. Cui, X. Duan, J. Hu, and C. M. Lieber, "Doping and electrical transport in silicon nanowires," *The Journal of Physical Chemistry B*, (June 2000), vol. 104, no. 22, pp. 5213–5216,.
- [3] Z. Chen, J. Appenzeller, Y. Lin, J. Sippel-Oakley, A. G. Rinzler, J. Tang, S. J. Wind, P. M. Solomon, and P. Avouris, "An integrated logic circuit assembled on a single carbon nanotube," *Science*, (Mar. 2006), vol. 311, no. 5768, p. 1735.
- [4] P. D. Tougaw and C. S. Lent, "Logical devices implemented using quantum cellular automata," *Journal of Applied Physics*, (Feb. 1994), vol. 75, pp. 1818– 1825.
- [5] S. Khasanvis et al., "Hybrid graphene nano-ribbon CMOS tunneling volatile memory fabric," *Proceedings of the 2011 IEEE/ACM International Symposium on Nanoscale Architectures*, San Diego, CA, (June 2011), pp. 189-195.
- [6] P. Shabadi et al., "Towards logic functions as the device," *Proceedings of the* 2010 IEEE/ACM International Symposium on Nanoscale Architectures (NANOARCH), (June 2010), pp. 11-16.
- [7] C. P. Collier, E. W. Wong, M. Belohradsk, F. M. Raymo, J. F. Stoddart, P. J. Kuekes, R. S. Williams, and J. R. Heath, "Electronically configurable Molecular-Based logic gates," *Science*, (Jul. 1999), vol. 285, no. 5426, pp. 391–394.
- [8] A. Schmid and Y. Leblebici, "A highly fault tolerant PLA architecture for failure-prone nanometer CMOS and novel quantum device technologies," *Proceedings of the 19th IEEE International Symposium on Defect and Fault Tolerance in VLSI Systems*, (Oct. 2004), pp. 39 – 47.
- [9] M. M. Khan, P. Narayanan, P. Vijayakumar, I. Koren, C. M. Krishna, C. A. Moritz, "Biased Voting for Improved Yield in Nanoscale Fabrics," Proceedings of IEEE International Symposium on Defect and Fault Tolerance (DFT 2011), (Oct 2011), pp. 79-85.
- [10] M. M. Khan, P. Narayanan, P. Vijayakumar, I. Koren, C. Krishna, "FastTrack: Toward nanoscale fault masking with high performance," *IEEE Transactions on Nanotechnology*, (July 2012), vol. 11, issue 4, pp. 720-730.

- [11] C. A. Moritz, P. Narayanan, and C. O. Chui, "Nanoscale application specific integrated circuits," in *Nanoelectronic Circuit Design, N. K. Jha and D. Chen, Eds. Springer New York*, 2011, pp. 215–275.
- [12] T. Wang, P. Narayanan, and C. A. Moritz, "Heterogeneous Two-Level Logic and Its Density and Fault Tolerance Implications in Nanoscale Fabrics," *IEEE Transactions on Nanotechnology*, (Jan. 2009), vol. 8, no. 1, pp. 22-30.
- [13] C. Moritz, T. Wang, P. Narayanan, M. Leuchtenburg, Y. Guo, C. Dezan, and M. Bennaser, "Fault-tolerant nanoscale processors on semiconductor nanowire grids," *IEEE Transactions on Circuits and Systems I*, (Nov. 2007), vol. 54, no. 11, pp. 2422 –2437.
- [14] P. Narayanan, M. Leuchtenburg, T. Wang, and C. A. Moritz, "CMOS control enabled Single-Type FET NASIC," *Proceedings of the 2008 IEEE Computer Society Annual Symposium on VLSI*, (April 2008), pp. 191–196.
- [15] T. Wang, P. Narayanan, and C. A. Moritz, "Combining 2-level logic families in grid-based nanoscale fabrics," *Proceedings of the 2007 IEEE International Symposium on Nanoscale Architectures*, San Jose, CA, (Oct. 2007), pp. 101-108.
- [16] P. Narayanan, C. A. Moritz, K. W. Park, Chi On Chui, "Validating cascading of crossbar circuits with an integrated device-circuit exploration," *Proceedings of the 2009 IEEE International Symposium on Nanoscale Architectures*, (July 2009), pp. 37-42.
- [17] T. Wang, M. Bennaser, Y. Guo and C. A. Moritz, "Wire-Streaming Processors on 2-D Nanowire Fabrics", in *Proceedings of Nanotech 2005, Nano Science and Technology Institute, Anaheim*, 2005.
- [18] P. Narayanan, T. Wang, M. Leuchtenburg, C. A. Moritz, "Image Processing Architecture for Semiconductor Nanowire based Fabrics," *Proceedings of IEEE* 8th International Conference on Nanotechnology, Arlington, (Aug. 2008), pp. 677-680.
- [19] P. Narayanan, J. Kina, P. Panchapakeshan, C. O. Chui and C. A. Moritz, "Integrated device-fabric exploration and noise mitigation in nanoscale fabrics," *IEEE Transactions on Nanotechnology*, (July 2012), vol. 11, issue 4, pp. 687-700.
- [20] Y. Yellambalase, M. Choi, Y. Kim, "Inherited Redundancy and Configurability Utilization for Repairing Nanowire Crossbars with Clustered Defects," in *Proc. IEEE 25th International Symposium on Defect and Fault Tolerance in VLSI Systems (DFT)*, Arlington, (Oct. 2006), pp. 98-106.

- [21] Y. Dotan, N. Levison, R. Avidan, D. K. Lilja, "History Index of Correct Computation for Fault-Tolerant Nano-Computing," *IEEE Transactions of Very Large Scale Integration (VLSI) Systems*, (July 2009), pp. 943-952.
- [22] S. C. Goldstein, M. Budiu, "NanoFabrics: spatial computing using molecular electronics," 28th Annual International Symposium on Computer Architecture, Goteburg, (2001), pp. 178-189.
- [23] A. KleinOsowski et al, "Exploring Fine-Grained Fault Tolerance for Nanotechnology Devices With the Recursive NanoBox Processor Grid," *IEEE Transactions on Nanotechnology*, (Sep. 2006), pp. 575-586.
- [24] M. Tehranipoor, "Defect tolerance for molecular electronics-based nanofabrics using built-in self-test procedure," in *Proc. IEEE 20th International Symposium on Defect and Fault Tolerance (DFT)*, Monterey, (Oct. 2005), pp. 305-313.
- [25] J. G. Brown, R. D. Blanton, "CAEN-BIST: testing the nanofabric," International Test Conference (ITC), Charlotte, (Oct. 2004), pp. 462-471.
- [26] W. Rao, A. Orailoglu, R. Karri, "Logic Mapping in Crossbar-Based Nanoarchitectures," *IEEE Design & Test of Computers*, (Feb 2009), pp 68-77.
- [27] T. Hogg, G. S. Snider, "Defect-tolerant adder circuits with nanoscale crossbars," *IEEE Transactions on Nanotechnology*, (Mar. 2006), pp 97-100.
- [28] A. DeHon, H. Naeimi, "Seven strategies for tolerating highly defective fabrication," *IEEE Design & Test of Computers*, (Aug 2005), pp 306-315.
- [29] J. von Neumann, "Probabilistic logics and the synthesis of reliable organisms from unreliable components," *Automata Studies*, C. E. Shannon and J. McCarthy. Princeton, NJ: Princeton Univ. Press, (1956), pp. 43–98.
- [30] J. Sklaroff, "Redundancy management technique for space shuttle computers," *IBM J. Res. Dev.*, (Jan. 1976), vol. 20, no. 1, pp. 20–28.
- [31] V. Julien, B. Alberto and G. Patrick. "Using TMR architectures for yield improvement," in *Proceedings of IEEE International Symposium. on Defect and Fault Tolerance in VLSI systems, (DFTVS)*, Boston, (Oct. 2008). pp. 7–15.
- [32] M. Hunger and S. Hellebrand, "The impact of manufacturing defects on the fault tolerance of TMR-systems," in *Proc. IEEE 25th Int. Symp. on Defect and Fault Tolerance in VLSI Systems (DFT)*, Kyoto, (Oct. 2010), pp 101-108.
- [33] I. Koren and C. M. Krishna, Fault-Tolerant Systems, San Fransisco, CA: *Morgan Kaufmann Publishers*, (2007).

- [34] T. Ban, L. A. de Barros Naviner, "A simple fault-tolerant digital voter circuit in TMR nanoarchitectures," *Proceedings of 8th IEEE International NEWCAS Conference*, Montreal, (June 2010), pp 269-272.
- [35] K. Granhaug and S. Aunet, "Improving yield and defect tolerance in subthreshold CMOS through output-wired redundancy," J. Electron. Testing, (Jun. 2008), vol. 24, no. 1–3, pp. 157–163.
- [36] A. Namazi, M. Nourani and M. Saquib, "A Fault-Tolerant Interconnect Mechanism for NMR nanoarchitectures," *IEEE Trans. Very Large Scale Integration (VLSI) Systems*, (Oct. 2010), pp 1433-1446.
- [37] C. A. Moritz, T. Wang, P. Narayanan, M. Leuchtenburg, Y. Guo, C. Dezan, and M. Bennaser, "Fault-tolerant nanoscale processors on semiconductor nanowire grids," *IEEE Transactions on Circuits and Systems I*, (Nov. 2007), vol. 54, no. 11, pp. 2422 –2437.
- [38] S. Roy and V. Beiu, "Majority multiplexing-economical redundant fault-tolerant designs for nanoarchitectures," *IEEE Transactions of Nanotechnology*, (Jul. 2005), vol. 4, no. 4, pp. 441–451.
- [39] J. Han, J. Gao, P. Jonker, Y. Qi, and J. A. B. Fortes, "Toward hardwareredundant, fault-tolerant logic for nanoelectronics," *IEEE Design & Test Computers*, (Jul-Aug. 2005), vol. 22, no. 4, pp. 328–339.
- [40] J. Han and P. Jonker, "A fault-tolerant technique for nanocomputers: NAND multiplexing," *Proceedings of Annual Advanced School for Computing and Imaging Conference*, Lochem, (2002), pp. 59–66.
- [41] P. Narayanan, K. W. Park, C. O. Chui, and C. Moritz, "Manufacturing pathway and associated challenges for nanoscale computational systems," *Proceedings of the 9th IEEE Conference on Nanotechnology*, Genoa, (July 2009), pp. 119 – 122.
- [42] E. R. Hnatek, Integrated Circuit Quality and Reliability, New York: *Marcel Dekker Inc.*, (1995).
- [43] I. Koren, "Analysis of the Signal Reliability Measure and an Evaluation Procedure," *IEEE Transactions on Computers*, (March 1979), pp. 244-249.
- [44] D.T. Franco, M.C. Vasconcelos, L. Naviner, J.F. Naviner, "Reliability analysis of logic circuits based on signal probability," *Proceedings of the 2008 IEEE International conference on electronics, circuits and systems*, (Sept 2008), pp 670-673.